Author's response to reviews

Title: Investigating the effect of intra-operative infiltration with local anaesthesia on the development of chronic postoperative pain after inguinal hernia repair. A randomized placebo controlled triple blindered and group sequential study design [NCT00484731]

Authors:

Philipp Honigmann (philipp.honigmann@ksl.ch)
Henning Fischer (henning.fischer@ksl.ch)
Anita Kurmann (anita.kurmann.1@ksl.ch)
Laurent Audige (laurent.audige@aofoundation.org)
Guido Schuepfer (guido.schuepfer@ksl.ch)
Juerg Metzger (juerg.metzger@ksl.ch)

Version: 2 Date: 1 November 2007

Author's response to reviews: see over
Author's response to reviews

Title:
Investigating the effect of intra-operative infiltration with local anaesthesia on the development of chronic postoperative pain after inguinal hernia repair. A randomized placebo controlled triple blinded and group sequential study design [NCT00484731]

Authors:
Philipp Honigmann
Henning Fischer
Anita Kurmann
Laurent Audigé
Guido Schüpfer
Jürg Metzger

The following change were made in the submitted manuscript:

1. Title – full stop removed
2. Author list – full forenames provided
4. Abstract & main section – Trial registration deleted in main section and moved to abstract
4. Competing interests – ‘author(s)’ changed to ‘authors’
5. Discussion – www.pubmed.org removed from text and inserted as citation [36]
6. Table placement text removed
7. Small table removed from page 10 and inserted as full sentence into text
8. Table moved after ‘References’
9. Figures removed from the manuscript
10. Tables inserted as Excel-tables
11. Title of tables moved above the table